**Author's response to reviews**

**Title:** Joint hypermobility in children with idiopathic scoliosis: SOSORT award 2011 winner

**Authors:**

Dariusz Czaprowski (dariusz.czaprowski@interia.pl)
Tomasz Kotwicki (tomaszkotwicki@poczta.onet.pl)
Paulina Pawlowska (wloska22@interia.pl)
Lukasz Stolinski (stolinskilukasz@op.pl)

**Version:** 3  **Date:** 1 October 2011

**Author's response to reviews:**

Paper “Joint hypermobility in children with idiopathic scoliosis: SOSORT award 2011 winner” written by Dariusz Czaprowski, Tomasz Kotwicki, Paulina Pawlowska, Lukasz Stolinski was changed according to the following suggestions:

1. We removed the keywords from the submissions system version of the abstract.
2. Table 12 has been placed in the manuscript following the references.
3. We have done References improvement.
4. The telephone/ fax numbers and email addresses have been removed from the affiliations.
5. All vertical lines from tables have been removed.
6. All figure legends were placed together before the references.